Author’s response to reviews

Title: A Fast Healthcare Interoperability Resources (FHIR) Layer Implemented Over i2b2

Authors:
Abdelali Boussadi (abdelali.boussadi@aphp.fr; boussadiabdelali@yahoo.fr)
Eric Zapletal (eric.zapletal@aphp.fr)

Version: 3 Date: 24 Jul 2017

Author’s response to reviews:

Dear editor,

We have redesigned Figure 5 as two of the icons used previously are not freely reusable. We used two other icons which are under Creative Commons CC0 license, downloaded here:

Icon 1: https://pixabay.com/p-158474/?no_redirect
Icon 2: https://pixabay.com/p-309582/?no_redirect

We described this in the legend of the figure. We designed the other parts of Figure 5.

According to your comments we have also added three new sections:

• Acknowledgments: in page 19
• Ethics and clinical data: in page 19 and
• Availability of data and material: in page 20.

Changes brought to the text are highlighted in yellow in the article.

Abdelali Boussadi, PhD.